Author's response to reviews

Title: Variable expressivity of FGF3 mutations associated with deafness and LAMM syndrome

Authors:

Saima Riazuddin (saima.riazuddin@cchmc.org)
Zubair M. Ahmed (zubair.Ahmed@cchmc.org)
Rashmi S. Hegde (rashmi.hegde@cchmc.org)
Shaheen N. Khan (snkhan52@hotmail.com)
Idrees Nasir (idreesnasir@cemb.edu.pk)
Uzma Shaukat (uzmashaukat283@hotmail.com)
Sheikh Riazuddin (riaz@lhr.comsats.net.pk)
John A. Butman (JButmanA@cc.nih.gov)
Andrew J. Griffith (griffita@nidcd.nih.gov)
Thomas B. Friedman (friedman@nidcd.nih.gov)
Byung Yoon Choi (twinwifegirl@empal.com)

Version: 7 Date: 11 January 2011

Author's response to reviews:

January 12th, 2011

Dear Ms Martin, for Jenny Leighon behalf of Professor Wasim Ahmad

We have made changes that you request.

We thank you again for your prompt decision on our manuscript.

Sincerely,

Byung Yoon Choi, MD, PhD
Assitant professor
Department of Otorhinolaryngology
Seoul National University College of Medicine
Seoul National University Bundang Hospital, Seongnam, Korea